Author's response to reviews

Title: Development of a validation algorithm for 'present on admission' flagging

Authors:

Terri J Jackson (t.jackson@uq.edu.au)
Jude L Michel (judith.michel@uq.edu.au)
Rosemary Roberts (rroberts@aapt.net.au)
Jennie Shepheard (Jennie.Shepheard@dhs.vic.gov.au)
Diana Cheng (D.Cheng@latrobe.edu.au)
Julie Rust (J.Rust@usyd.edu.au)
Catherine Perry (Catherine.Perry@dhs.vic.gov.au)

Version: 9 Date: 23 November 2009

Author's response to reviews:

small revisions required:

p 4: remove 'episode' in sentence where references [8-10] appears

p11: new para required beginning at 'We envisage...'

Endnote 23: remove excess '.' (period or full stop) in reference